Author's response to reviews

Title: Are GPs adequately equipped with the knowledge for educating and counseling of families with ADHD children?

Authors:

Ahmad Ghanizadeh (ghanizad@sina.tums.ac.ir)
Nabi Zarei (sdd@as-da.ir)

Version: 3 Date: 1 January 2010

Author's response to reviews:

Dear Nina Titmus

Thank you for your email about manuscript (MS: 7010666312492338). The manuscript was revised according to the comment. The sentence of "Just 20% agreed that ADHD is not a serious problem" was deleted from the results of the abstract.

Sincerely
Ahmad Ghanizadeh